Author’s response to reviews

Title: A Bayesian Decision Support Sequential Model for Severity of Illness Predictors and Intensive Care Admissions in Pneumonia Short/ Running Title: Bayesian Pneumonia Decision

Authors:

Amado Baez (aabaezmd@gmail.com)
Laila Cochon (lailacochn@hotmail.com)
Jose Nicolas (nicolas@ub.edu)

Version: 3 Date: 20 Dec 2019

Author’s response to reviews:

1 - Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

ADDED PAGE 17

2 - Please include an Availability of Data and Materials statement in the Declarations and ensure it reflects accurately one of the formats indicated in our submission guidelines for the raw data: https://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials

ADDED Page 17

3 - At this stage, please upload your proofread manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colours. All relevant tables and figures should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Should you wish to respond to these revision requests, please include the information in the designated input box only.

UPLOADED